Author’s response to reviews

Title: Hepatoprotection with chloroform extract of Launaea procumbens against CCl4 induced injuries in rat

Authors:

Rahmat Ali Khan i (rahmatgul_81@yahoo.com)

Version: 3 Date: 29 September 2011

Author’s response to reviews: see over
Response to reviewer comments

Editor,

BMC Complementary and Alternative Medicine

Dear Sir,

We are submitting a revised manuscript entitled, “Hepatoprotection with chloroform extract of Launaea procumbens against CCl₄ induced injuries in rat” in your esteemed journal. We improved the comments raised by the reviewer in manuscript and also given point by point. We hope that you will consider our paper for publication in your journal.

Reviewer 1 Prof. Dr. Suh-Ching Yang

Dear Sir

I am very pleased for your kind comments on our manuscript. Sir I have corrected the manuscript according to your suggestion and hope that this time, our manuscript will be able to publish. Please check point by point response to

1. Page 2, in Abstract, high performance liquid chromatography (HLPC)” is corrected.
2. Page 5, hepatoprotective potential through the antioxidative property of LPCE is modified.
3. At page 6 no 2.4 is corrected.
4. Page 7 “HDL” and “LDL” are replaced with HDL-cholesterol” and “LDL-cholesterol”.
5. Page14, in 3.3 corrections is carried out.
6. Page 15, in 3.5 described properly.
7. In title of Table 5, “H₂O₂” is deleted.
8. LPCE in Figure 3 and Figure 4 dose of LPCE is 200 mg/kg b.w.
9. Magnification of Figure 4 is adjusted.
Reviewer 2

Prof. Dr. Omar M Abdel Salam

Dear Sir

I am very pleased for your kind comments on our manuscript. Sir I have corrected the manuscript according to your suggestion and hope that this time, our manuscript will be able to publish. Please check point by point response to

1. CCl₄ is given twice a week for four weeks, mentioned in the manuscript.

2. First para of the manuscript is modified as per given comments.

Thanks once again for your guidance and remarks
Regards and best wishes

Dr. Rahmat Ali Khan
Chairman
Department of Biotechnology
UST Bannu, KPK, Pakistan